Author’s response to reviews

Title: High-fat diet aggravates postoperative cognitive dysfunction in aged mice

Authors:

Lan Wei (371157467@qq.com)
Minmin Yao (yao.minmin@zs-hospital.sh.cn)
Zhimeng Zhao (474385823@qq.com)
Hui Jiang (13917351884@163.com)
Shengjin Ge (ge.shengjin@fudan.edu.cn)

Version: 2 Date: 19 Jan 2018

Author’s response to reviews:

BMC Anesthesiology
No. BANE-D-17-00396 R2

Title: High-fat diet aggravates postoperative cognitive dysfunction in aged mice

Dear editors:

We would like to thank you for giving us constructive suggestions which have help us in depth to improve the quality of the paper. Here we submit the revised manuscript, which has been revised according to the editors’ suggestions. And we have provide the reference number for the ethics approval in Page 16 line 7-8.

Thank you very much for the valuable suggestions about our manuscript. We have carefully revised the manuscript, and hope that the corrections will meet with approval.

Yours sincerely,

ShengJin Ge

January 19, 2018